Author's response to reviews

Title: Longitudinal monitoring of Ehrlichia ruminantium infection in Gambian lambs and kids by pCS20 PCR and MAP1-B ELISA

Authors:

Bonto Faburay (b.fabureh1@vet.uu.nl)
Dirk Geysen (dgeysen@itg.be)
Susanne Munstermann (smunster@freenet.de)
Lesley Bell-Sakyi (Isakyi@staffmail.ed.ac.uk)
Frans Jongejan (F.Jongejan@vet.uu.nl)

Version: 5 Date: 23 July 2007

Author's response to reviews:

Dear Correction team,

The shades in the Tables have been removed; instead animal numbers are converted to bold for mortality cases. This is indicated in the Keys.

The Figure is in Word format and I hope it conforms to your requirements.

All Tables are numbered in consecutive order as they appear in the text.

Best wishes,

Dr. Bonto Faburay